Author's response to reviews

Title: Expansion of seasonal influenza vaccination in the Americas

Authors:

Alba María Ropero-Álvarez (roperoal@paho.org)
Hannah J Kurtis (kurtisha@paho.org)
M. Carolina Danovaro-Holliday (danovarc@paho.org)
Cuauhtémoc Ruiz-Matus (ruizcuaau@paho.org)
Jon K Andrus (andrusjo@paho.org)

Version: 6 Date: 17 September 2009

Author's response to reviews: see over
17 September 2009
RE: Proofreading requested

Attention: Melissa Norton, M.D.
Editor-in-Chief, BMC Public Health

Dear Dr. Norton:

Thank you for giving us the opportunity to publish our manuscript “Expansion of seasonal influenza vaccination in the Americas,” with BMC Public Health. We have now reviewed the manuscript and we have made several small typographical changes. Additionally, in the footnotes below Table 1, the footnote “with chronic diseases” had been repeated multiple times and each time it had been given a new corresponding letter; this unnecessary replication has been corrected so that there is now only one footnote which is referenced multiple times in the table. We feel that this adjustment simplifies the footnotes, while conserving all information.

Again, thank you very much for this opportunity.

Sincerely yours,

Alba María Ropero-Álvarez